Author's response to reviews

Title: Increased tartrate-resistant acid phosphatase (TRAP) expression in malignant breast, ovarian and melanoma tissue: an investigational study

Authors:

Arnd Honig (arnd_hoenig@hotmail.com)
Lorenz Rieger (lrieger@gmx.de)
Michaela Kapp (M.Kapp@mail.uni-wuerzburg.de)
Mathias Krockenberger (matze.krock@web.de)
Matthias Eck (Matth.Eck@gmx.de)
Johannes Dietl (frauenklinik@mail.uni-wuerzburg.de)
Ulrike Kammerer (FRAK057@mail.UNI-wuerzburg.de)

Version: 6 Date: 18 July 2006

Author's response to reviews:

Dear Madam or Sir,

Formal corrections are now included, figures are cropped and the source of funding described.
Thank you very much in advance.

Arnd Honig